Author’s response to reviews

Title: Head and neck cancer survivors’ preferences for and evaluations of a post-treatment dietary intervention

Authors:
Sylvia Crowder (sylviac2@illinois.edu)
Katherine Douglas (kgd2@illinois.edu)
Andrew Fruge (fruge@auburn.edu)
William Carroll (wcarroll@uabmc.edu)
Sharon Spencer (sspencer@uabmc.edu)
Julie Locher (jlocher@uabmc.edu)
Wendy Demark-Wahnefried (demark@uab.edu)
Laura Rogers (rogersl@uab.edu)
Anna Arthur (aarthur@illinois.edu)

Version: 2 Date: 22 Aug 2019

Author’s response to reviews:

Department of Food Science and Human Nutrition
THE UNIVERSITY OF ILLINOIS
COLLEGE OF ACES
905 S GOODWIN AVENUE
URBANA, ILLINOIS 61801 August 22, 2019

Holly L Nicastro, Editor-in-Chief
Dear Dr. Nicastro,

Thank you for the opportunity to provide a revision to our manuscript entitled “Head and neck cancer survivors’ preferences for and evaluations of a post-treatment dietary intervention” which my co-authors and I are pleased to resubmit for publication in Nutrition Journal. The revised article describes our original work, and it has not been submitted for publication elsewhere.

We have made revisions to our manuscript based on the valuable suggestions from Reviewer 2 and believe it is now ready for publication. In order to expedite your decision, we have highlighted the revised content in red. Below please find point-by-point responses to the reviewer comments.

Reviewer 2 Comments for the Author:

The authors have addressed all concerns. A few minor comments to assist with clarity/readability:

1. Table 3 looks great. Very clear. Remember to refer to Table 3 in text.
Response: Table 3 is now referenced in-text on Page 8, Line 206.

2. Consider aligning your description with the labels provided in Table 3. Eg. "somewhat agree" and 'strongly agree' responses were collapsed into one category titled "agree".
Response: The description has been aligned with the labels provided in Table 3 and the results section has been updated to reflect the new categories on Page 7, Line 172-175 and Pages 8 and 9, Line 208-215.

Sincerely,

Anna E. Arthur, PhD, MPH, RD
Assistant Professor, Department of Food Science and Human Nutrition
University of Illinois
(217) 300 – 3014
aarthur@illinois.edu